Author’s response to reviews

Title: Exposure to artificial UV radiation in France: a population based study.

Authors:

Tarik Benmarhnia (benmarhnia.tarik@ouranos.ca)
Christophe Léon (christophe.leon@inpes.sante.fr)
François Beck (francois.beck@inpes.sante.fr)

Version: 2 Date: 17 October 2012

Author’s response to reviews:

The revisions done in this version correspond to an editorial request concerning:

Ethical Approval
Participant Consent
Title Page
Additional Sections (including the following sections: Competing Interests, Authors’ Contributions and Acknowledgements)
Line Numbers